Author’s response to reviews

Title: A Puzzling Case of Pylephlebitis Presenting as Spontaneous Coronary Sinus Thrombosis

Authors:
Michael Hart (michael.hart@hcmed.org)
Mengistu Simegn (Mengistu.simegn@hcmed.org)

Version: 2 Date: 25 Aug 2017

Author’s response to reviews:

Reviewer’s Comments

The presentation is inadequate and more details are needed:
Give detailed past medical, smoking, alcohol consumption etc, social and occupational history with dates.

-Now included in Case Summary section

What medications (doses and duration ) was she receiving on presentation, hospitalization and discharge.
-Now included in Case Summary section

Give detailed physical and neurological examination.
-Now included in Case Summary section

Give a long term followup for at least 6 months.
-Now included in Case Summary section

Start the discussion with description of what is unique to this patient compared to the literature
-Amended Discussion section to reflect this